Author's response to reviews

Title: Addressing maternal healthcare through demand side financial incentives: experience of Janani Suraksha Yojana Program in India

Authors:

Saji S Gopalan (sajisaraswathyg@gmail.com)
Varatharajan D (sajisaraswathyg@yahoo.com)

Version: 5 Date: 22 August 2012

Author's response to reviews:

MS: 9240063386588957
Authors' response

1. Language Editing
Authors' response: We have done a rigorous editing of the manuscript externally and have made corrections (indicated in track changes).

Hope this will enable further processing of the manuscript.

Sincerely,

S S Gopalan